Author's response to reviews

Title: Improving the implementation of a hand hygiene intervention: Using psychological theory

Authors:

Veronique M Boscart (boscart.veronique@torontorehab.on.ca)
Geoff R Fernie (fernie.geoff@torontorehab.on.ca)
Jae H Lee (jhl.lee@utoronto.ca)
Susan B Jaglal (susan.jaglal@utoronto.ca)

Version: 2 Date: 13 October 2011

Author's response to reviews:

Dear,

The following required editorial revisions have been made to the manuscript:

- The title page is now included as page 1. The title page includes the title, authors, author affiliations and corresponding author details.
- A competing interests section is included at the end of the manuscript.
- The Authors’ Contributions section is included at the end of the manuscript.

We look forward to your review of the manuscript.

Veronique Boscart